Author's response to reviews

Title: Evaluation of antimalarial activity of leaves of Acokanthera schimperi and Croton macrostachyus against Plasmodium berghei in Swiss albino mice

Authors:

Tigist Mohammed (tgdeg43@yahoo.com)
Berhanu Erko (berhanue@yahoo.com)
Mirutse Giday (mirutseg@yahoo.com)

Version: 2 Date: 4 February 2014

Author's response to reviews:

Journal Editorial Office,

Thank you for the comments which we have tried to address.

1. We have included some references in the background section to justify the testing of our extracts. We have now incorporated results of previous in vitro and in vivo antiplasmodial studies conducted on the plants.

2. We have moved the tables to the end of the manuscript.

Regards,

Mirutse Giday, PhD
Corresponding Author